Author's response to reviews

Title: A European project on incidence, treatment, and outcome of sarcoma

Authors:

Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)
Emanuela Fadda (emanueal.fadda@unipd.it)
Luca Cegolon (mrcegolon@hotmail.com)
Maria C Montesco (maria.montesco@sanita.pd.it)
Isabel Ray-Coquard (drcegolon@yahoo.com)
Alessandra Buja (alessandra.buja@unipd.it)
Ugo Fedeli (ugo.fedeli@ulssasolo.ven.it)
Alvise Frasson (alvise.frasson@unipd.it)
Paolo Spolaore (paolo.spolaore@ulssasolo.ven.it)
Carlo R Rossi (carlor.rossi@unipd.it)

Version: 2 Date: 19 March 2010

Author's response to reviews:

Dear BMC Editorial Team,

Thank you for your email.

A new manuscript has now been uploaded with the two additional requested points at the end of the main body text:

- Competing Interests
- Authors' contributions

It has been a pleasure to work with you and look forward to the paper being published.

Best Wishes.

Dr. Luca Cegolon, MD, FFPH
Corresponding Author